Author's response to reviews

Title: The opportunities for and obstacles against prevention: the example of Germany in the areas of tobacco and alcohol

Authors:

Ulla Walter (Walter.Ulla@MH-Hannover.de)
Marc Suhrcke (m.suhrcke@uea.ac.uk)
Miriam G Gerlich (Gerlich.Miriam@MH-Hannover.de)
Till A Boluarte (till.boluarte@uni-wh.de)

Version: 5 Date: 17 August 2010

Author's response to reviews: see over
Resubmission for BMC Public Health after reformatting
Category “Debate”.

Ms. No.: 6236362423490892

To the Editors
We would like to thank the editorial office for informing us on necessary format changes in the article “The opportunities for and obstacles against prevention: the example of Germany in the areas of tobacco and alcohol”. Please see below how we responded to the points raised.

Formatting changes to be made:
-----------------------------

Major revisions (we require the author to make these changes)

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1.
Please do not include vertical lines in your tables.
We removed the tables and changed them to figures, which we submitted as separate files. We changed the corresponding links within the text.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Keywords: please remove the keywords section from the end of the abstract in the submission system.
We removed the Keywords.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We proof read the manuscript and corrected minor typographical/formatting errors.